Author's response to reviews

Title: Health service utilization after terrorism: A longitudinal study of survivors of the 2011 Utøya attack in Norway

Authors:

Lise E Stene (lise.stene@nkvts.unirand.no)
Grete Dyb (grete.dyb@nkvts.unirand.no)

Version: 3 Date: 22 August 2014

Author's response to reviews:

Dear editorial board of the BMC Health Services Research,

Thank you for your e-mail concerning our manuscript “Health service utilization after terrorism: A longitudinal study of survivors of the 2011 Utøya attack in Norway”. We have revised the manuscript in accordance with your remarks.

1. We have completed a declaration of competing interests.

2. The following ethics statement is included in the Method section, page 6, lines 3-5:

“Participants gave written informed consent and the Regional Committee for Medical and Health Research Ethics approved the study.”

With respect to the availability of supporting data, we cannot openly share data as it would breach participant confidentiality.

3. The revised manuscript includes both line and page numbers.

Best regards,
Lise Eilin Stene